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Taking Data

« On Feb 6, a FEB chain caused an error which prevents the
DAQ from running. We went to the MINOS Hall to determine
what was wrong. A FEB in that chain had only one green light
on instead of 2. This FEB did not previously show a problem.
This FEB was replaced and this fixed the problem

 One of our control room PCs failed over the weekend. The
PPD FEF group is looking into the problem. With 4 control
room computers, we moved the displays to another computer.



Filling the He Target

The He target is full.

The target absolute pressure & temperature is stable. The pressure
IS the same as it was during the LE Run

The He target is ready for He Target data taking. In fact we may
decide the He target as fine for He target data taking before the
temperature is stabilized.

— Note that before we started filling the He target we were taking
empty target data which is necessary for background subtraction



Putting the Roof Back On

 Two of the 3 roof panels are on. These cover the MINERVA
detector. We plan putting on all 3 panels

— Uncovered are parts of the He Target and the veto counters.
One of the drip pans is over the veto counters.

— We plan on putting the roof on during next week’s downtime

— We would like to move the hook ~1/2” on one of the panels to
make it easier to connect the panel to the spreader bar.



Rock Muons/POT

-

ha
t

0.9

0.8

M

—
n
IIIIIIIIIIIIIIIIIIIIII

0.7

0.6

0.5

Protons per pulse / 10"
4
L]
L ]
‘.
+
Rock muons per 10" POT

0.3

L L 1 T T S SN S S S—
: P P D0 12-0712-1412-2112-2801-0401-1101-1801-2502-01

1
12-0712-1412-2112-2801-0401-1101-1801-2502-01 Date
Date

+

0.5

 Rock muons are muons generated in the berm, not the detector.

* Plot of rock muons/POT. The low region at the beginning of Dec was
horn off studies.

e Rock muons/POT Is constant



' MUON ALCOVE 4 W08
L HEL | UM- PANEL

Muon Monitor 4 Gas Rack

For the downtime next we need to
have AD install in the Muon Monitor 4
alcove the pressure transducer in the
gas line & the temperature sensor.

* Done by AD techs



Event Display
Outer Calorimeter
¢ \

\

I . o p
W e | | 1§ W e L] w e
5 | 5 ;
F £d F [
G\IIIIIIIIIIIIIIIIIIII\\II@\IIIIIIIIIIIIIIIIIIII\\II 0II\\ T
|
220 220 220
S : > I 2
ol X View Sl Y Miew o 3l)
80 80 80
410 L - L 60 60
o e o —
y : y || o Z N1ew
@\III | I I | L @\III | I I O | L DII\\ T Tl
[8Y] m -
i i e
g: ______g: g: |
¢ |8 2
0 ﬂl 0 0

% [5 % lll} 1% le} 2'5 3|l} 3%4|0 4% 5|l} 5% 6‘0 6% ?ID ?g 8|D 8% 9|D 9%15[1[5511[115 % [5 % 1|0 1% le} 2'5 3|l} 3%4|0 4% 5|l} 5% 6‘0 6% 710 ?g 8|D 8% 9|D 9%15[1[5511[115 E 6 % 1‘[; lg 2|G 2'5 3|G 3%4'0 4% 5|D SE E:b 6% fﬂ ?g Bb Bg Qb 9%1[5[1[5511[1{5

Neutral Current Event



NuMI| Beam Plots
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Protons for the Week

Week to 00:00 Monday 10 February 2014
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FY2014 Protons

FYI14 NuMI protons to 00:00 Monday 10 February 2014
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Profons per week (E18)
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